Check for updates

G OPEN ACCESS

Citation: Naylor-Leyland G, Collin SM, Gatluak F, den Boer M, Alves F, Mullahzada AW, et al. (2022) The increasing incidence of visceral leishmaniasis relapse in South Sudan: A retrospective analysis of field patient data from 2001–2018. PLoS Negl Trop Dis 16(8): e0010696. https://doi.org/10.1371/ journal.pntd.0010696

Editor: Mitali Chatterjee, Institute of Postgraduate Medical Education and Research, INDIA

Received: May 1, 2022

Accepted: July 23, 2022

Published: August 18, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0010696

Copyright: © 2022 Naylor-Leyland et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

RESEARCH ARTICLE

The increasing incidence of visceral leishmaniasis relapse in South Sudan: A retrospective analysis of field patient data from 2001–2018

Gabriel Naylor-Leyland¹, Simon M. Collin^{2,3}, Francis Gatluak⁴, Margriet den Boer⁵, Fabiana Alves⁶, Abdul Wasay Mullahzada¹, Koert Ritmeijer¹*

1 Médecins Sans Frontières, Amsterdam, The Netherlands, 2 UK Health Security Agency (UKHSA), London, United Kingdom, 3 Departamento de Medicina Social, Universidade Federal do Espírito Santo, Vitória, Brazil, 4 Médecins Sans Frontières, Lankien, South Sudan, 5 Médecins Sans Frontières, London, United Kingdom, 6 Drugs for Neglected Diseases *initiative* (DNDi), Geneva, Switzerland

* koert.ritmeijer@amsterdam.msf.org

Abstract

Background

Visceral Leishmaniasis (VL) is endemic in South Sudan, manifesting periodically in major outbreaks. Provision of treatment during endemic periods and as an emergency response is impeded by instability and conflict. Médecins Sans Frontières (MSF) has provided health care in South Sudan since the late 1980's, including treatment for 67,000 VL patients. In recent years, MSF monitoring data have indicated increasing numbers of VL relapse cases. A retrospective analysis of these data was performed in order to provide insight into the possible causes of this increase.

Methodology/Principal findings

Programme monitoring data from the MSF hospital in Lankien, Jonglei State, South Sudan, for the period 2001–2018 were analysed to detect trends in VL relapse as a proportion of all VL cases presenting to MSF treatment centres. Routinely collected patient-level data from relapse and primary VL cases treated at all MSF sites in South Sudan over the same period were analysed to describe patient characteristics and treatments received. VL relapse as a proportion of all VL cases increased by 6.5% per annum (95% CI 0.3% to 13.0%, p = 0.04), from 5.2% during 2001–2003 to 14.4% during 2016–2018. Primary VL and VL relapse patients had similar age, sex and anthropometric characteristics, the latter indicating high indices of undernutrition which were relatively constant over time. Clinical factors (Hb, spleen size, and VL severity score) also did not vary substantially over time. SSG/PM was the main treatment regimen from 2001–2018, used in 68.7% of primary and 70.9% of relapse VL cases; AmBisome was introduced in 2013, received by 22.5% of primary VL and 32.6% of VL relapse cases from 2013–2018.

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

Conclusion

Increasing incidence of VL relapse in South Sudan does not appear to be explained by changes in patient characteristics or other factors. Our data are concerning and may indicate an emergence of treatment-resistant parasite strains, decreasing the effectiveness of treatment regimens. This warrants further investigation as a causal factor. New chemical entities that will enable safe and highly effective short-course oral treatments for VL are urgently needed.

Author summary

Visceral leishmaniasis is a fatal parasitic disease which affects impoverished populations living in remote areas. It has caused very high mortality in South Sudan where several major outbreaks have occurred in the past 40 years. In East Africa, the disease can be successfully treated with a combination of injectable sodium stibogloconate and paromomycin given for 17 days. South Sudan was the first country where this regimen was introduced. Now, nearly 20 years later, our data show that it may have become less effective. In our clinic in Lankien, situated in the middle of the endemic area, more and more patients present with relapsed disease and need to be retreated. We have analysed our data and found that the relapse rate as a proportion of all VL cases has increased by 6.5% per annum since 2003. It is possible that this is caused by treatment-resistant parasite strains. Since there is currently no alternative first line treatment, this is of concern. We have planned a prospective observational study where we will follow patients for 12 months in order to determine the current relapse rate, and we highlight the need for new chemical entities that will enable safe and effective short-course oral treatments.

Introduction

Visceral leishmaniasis (VL, also known as 'kala azar') is a vector-borne disease that is 100% fatal if untreated. Substantial progress has been made in reducing the burden of VL in endemic south Asian countries, with reported cases in India, Bangladesh and Nepal declining from around 80,000 in 1992 to 3,000 in 2019 and elimination becoming a real prospect [1]. By contrast, progress in East Africa has been slow, particularly in countries affected by conflict and political instability, with South Sudan experiencing sustained endemicity and several major VL epidemics over the past 40 years. Médecins Sans Frontières (MSF) is an international medical humanitarian aid organization which has been present in South Sudan before, during and after the civil wars which led to its independence, and has treated 67,000 VL patients since the 1980's. Conflict-related challenges to VL control, including lack of infrastructure and clinical staff, internally-displaced populations, and malnutrition [2], have been compounded recently by the complete withdrawal of UK aid funds for VL control in 2020, the COVID-19 pandemic, and longer-term threats to supplies of diagnostic kits and drugs [3,4]. Pentavalent antimonials have been the mainstay of treatment against VL for many decades, and resistance to these, as already established on the Indian subcontinent by the early 1990s [5], and potentially to newer therapeutic agents such as miltefosine and paromomycin (PM) [6], presents another potential challenge to VL treatment and control.

To avoid monotherapy and shorten treatment duration, first-line 30-day sodium stibogluconate (SSG) monotherapy was phased out in South Sudan from 2003 onwards, replaced by a 17-day injectable combination regimen of SSG and PM [7–9]. Equivalent efficacy of SSG/PM compared to SSG monotherapy was proven in clinical trials in east Africa, with combination therapy showing 91% efficacy [9], and this regimen was adopted in WHO recommendations in 2010 [10] and South Sudan Ministry of Health guidelines in 2011 [11]. Liposomal amphotericin B (AmBisome, Gilead Sciences Inc.) was introduced in the late 90's [12], but its prohibitively high price limited its use to severely ill patients (patients unable to walk or sit up, and with severe weight loss, severe anaemia, severe super-infections, and other markers of poor prognosis). Negotiation of a reduced 'access price' meant that use of AmBisome was gradually expanded from 2012 onwards as second-line treatment and for patients with high intolerance to SSG, including pregnant women, the elderly, and HIV-co-infected patients. AmBisome has been provided as a donation by the manufacturer via WHO for these indications since 2012.

Disease relapse after initial successful treatment of primary VL in a small proportion of patients is a well-described phenomenon, with evidence from trials and field studies indicating a relapse rate of around 5% for SSG/PM combination therapy in East Africa [7,13,14]. Relapse may be attributed to ineffective cellular immunity after treatment related to immunosuppressive conditions such as HIV, TB, and malnutrition, and/or to inadequate treatment resulting in significant persistent parasitaemia despite initial clinical cure. In settings such as South Sudan where active patient follow-up is extremely difficult and not routinely performed, VL relapse rates are monitored passively by VL re-treatment admissions (re-admissions) as a fraction of the total number of VL admissions. In the past few years, this passive monitoring has indicated an apparent increase in re-treatment rates. The aims of the present study were to analyse programme data to determine trends in VL relapse rates and to use patient-level data to generate hypotheses which might explain any such trends.

Methods

Ethics statement

This research fulfilled the exemption criteria set by the MSF Ethical Review Board (ERB) (https://scienceportal.msf.org/assets/7002) for retrospective analysis of routinely-collected clinical data from pre-existing, established programs, and thus did not require MSF ERB review (https://scienceportal.msf.org/assets/6996). It was conducted with permission from the Medical Director of the MSF Operational Centre Amsterdam.

Study design, setting and population

Due to the dynamic epidemiology of VL in South Sudan, characterised by large fluctuations in case numbers between locations and from year to year, with focal and seasonal outbreaks, MSF has a flexible operational model that has included provision of treatment at more than 25 sites across South Sudan, in Unity, Jonglei, and Upper Nile states. VL patient numbers in MSF's facilities over the past 10 years haved averaged over a thousand patients per year, ranging from a few hundred to over 6000 patients yearly depending on the epidemiology. This retrospective analysis used routine clinical data collected mainly from a Médecins Sans Frontières (MSF) treatment facility in Lankien, Jonglei State, in the north eastern part of South Sudan. This facility has been run by MSF since 1993, providing primary outpatient and inpatient health care services, comprehensive maternity care, and treatment for TB, HIV, and tropical diseases including VL. The catchment area population is mainly pastoralist, practicing cattle husbandry and agriculture. The facility is located in an area with high VL endemicity and 50.2% of all VL patients treated by MSF during the study period were treated here, and

75% of VL patients treated during 2013–2018. Between 2013–2018, 66% of all reported VL cases in South Sudan were treated by MSF (50% in Lankien).

Access to the facility is often difficult because of poor road conditions, flooding during the rainy season, and insecurity.

Diagnosis and treatment of VL in South Sudan

Clinical management followed WHO and South Sudan Ministry of Health guidelines [10,11,15] as summarised below.

Diagnosis. All patients presenting with a history of fever of more than 2 weeks and splenomegaly and/or lymphadenopathy and/or wasting were considered VL suspects and underwent further diagnostic evaluations. Patients without prior VL treatment history (suspect primary VL) were screened using the rK39 rapid diagnostic test (IT-Leish, Bio-Rad laboratories, USA) with a positive result considered confirmatory. Those testing negative were screened with the leishmania direct agglutination test (DAT, Royal Tropical Institute, Amsterdam, The Netherlands), with a high titer (≥1:6400) considered confirmatory. Those with an intermediate DAT titer (1:800–1:3200) underwent tissue aspiration (spleen or lymph node), with a positive result confirming VL. Patients with prior VL treatment history (suspect relapse VL) underwent tissue aspiration (mostly splenic) with diagnosis by microscopy. Routine testing of all primary and relapse VL patients for HIV commenced in 2012.

Treatment. First line treatment was SSG (20 mg/kg) in combination with PM (15 mg/kg) given on an ambulatory basis over 17 days with daily intramuscular injections. Patients contra-indicated for SSG (pregnant women), with known poor tolerability of SSG (age >45 years, severe VL), and patients who did not respond to SSG were treated with AmBisome by 6 IV infusions of 5 mg/kg on alternate days. Relapse patients were treated either with SSG (20 mg/ kg for 30 days) in combination with PM (15 mg/kg for 17 days) or AmBisome (6 IV infusions of 5 mg/kg on alternate days) [16,17]. Patients with HIV/VL co-infection received AmBisome. Patients with TB/VL typically have other symptoms qualifying as severe VL and are therefore often treated with AmBisome. The drugs were from the same quality-controlled source during the entire period (paromomycin: Gland Pharma, India; SSG: Albert David, India; AmBisome: Gilead, USA).

Discharge. Patients with an increased risk of treatment failure or relapse, i.e. patients with a prior episode of VL, or patients with inadequate or doubtful clinical response, required a negative parasitological test-of-cure by aspirate microscopy to confirm cure (at 21 days for primary VL, at 30 days for VL relapse). All other patients were discharged without test-of-cure. Patients were not actively followed up but were advised to come back to the clinic in case symptoms reoccurred.

Treatment adherence and default. Treatment adherence was encouraged through incentives, with patients receiving food rations, PlumpyNut supplement and mosquito nets when they completed their course of injections. Doses of SSG, PM or AmBisome were checked independently by the programme supervisor after weighing the patient on admission and every week thereafter, and the number of injections and doses were recorded in treatment records. A defaulter protocol applied throughout the study period whereby patients who missed two consecutive visits were actively traced to ensure that they completed their course of treatment.

Patient-level data

Patient-level data were obtained from a pseudonymised archive of routinely-collected clinical data from all MSF treatment centres in South Sudan from 2001–2018. Demographic, diagnostic, treatment, and discharge information were handwritten on cards from which data were

double checked before subsequent entry into electronic databases, which were merged for the purpose of this analysis. Patient cards from various treatment sites were lost during periods of acute armed conflict when MSF staff had to be evacuated and health facilities were looted or burned and, on other occasions, stored patient cards were lost during flooding. For these reasons, some patient-level data were missing, especially from peripheral sites. Potentially duplicate records created by merging databases were identified by combining patient number, treatment site, sex, and date of admission.

A relapse case was defined as any patient who was recorded with a diagnosis of 'Relapse VL' on admission, based on the attending medical staff asking the patient whether they had previously been treated for VL more than 30 days before admission for the current VL episode.

We used all available patient-level data to analyse primary and relapse VL patient characteristics (rather than restricting only to patients treated in Lankien) to maximise the sample size and because treatment site wasn't routinely recorded in the first year of the study period. To simplify comparisons of patient characteristics over time, we aggregated patient-level data into 3-year periods (2001–2003, 2013–2015, and 2016–2018) except for the years 2004–2012, which saw relatively low numbers of VL patients. These periods also broadly correspond with the introduction of SSG/PM combination therapy (from 2003/2004) and AmBisome (from 2012).

Patient-level clinical data included weight-for-age Z-score for children aged ≤ 10 years, BMI-for-age Z-score for 11–19 year-olds, BMI for adults aged 20+ years, haemoglobin (Hb), spleen size (distance in centimetres from left costal margin to tip of spleen). A VL severity score was derived from risk factors for VL mortality: age, nutritional status (weight-for-height Z-score for children and adolescents aged <19 years, BMI for adults aged 19+ years), haemoglobin (Hb), and level of weakness. Patients with a risk score associated with >20% probability of death were classified as having 'severe' VL and received AmBisome, aggressive antibiotic treatment and nutritional care. Time-to-relapse was calculated for VL relapse patients whose discharge date for their primary VL episode was recorded in the database.

Monitoring data

Programme monitoring data include aggregate counts of cases by treatment site, diagnosis (primary VL, VL relapse, and post-kala azar dermal leishmaniasis (PKDL)) and outcome. For the purpose of analysing trends in relapse, we used programme monitoring data only from Lankien because it is the only treatment site that has provided VL services throughout the study period (except from 06/2004-06/2005 when staff were evacuated for security reasons) and it has the most consistent and continuous recording of VL diagnostic data (S1 Table).

Statistical methods

Trends in VL relapse as a proportion of all VL cases (primary and relapse) recorded in programme monitoring data from 2001–2018 were analysed using Joinpoint trend analysis software (Joinpoint Regression Program, Version 4.7.0.0—February 2019; Statistical Methodology and Applications Branch, Surveillance Research Program, National Cancer Institute, Bethesda MD, USA). Trends were estimated as the annual percentage change (APC) by fitting models linear on the log of the percentage VL relapse with minimum 0 joinpoints, maximum 3 joinpoints, minimum 2 observations between joinpoints, 4,500 Monte Carlo permutations, and α = 0.05 [18]. Differences in characteristics of primary and relapse VL cases from patient-level data over time were tested using Chi-squared tests for categorical variables and non-parametric K-sample test on the equality of medians, non-parametric test-for-trend, and Student's t test (if normally distributed) for continuous variables, all with α = 0.05. Logistic regression was used to estimate odds ratios for VL relapse by HIV and TB status, adjusted for age and sex.

Results

Trends in VL relapse from MSF programme monitoring data

The proportions of all VL patients in Lankien who were diagnosed as primary VL or relapse VL for each year from 2001 to 2018 are summarised in Table 1. Join-point analyses indicated that the best-fit model had zero join points and an annual percent change in relapse as a proportion of all VL cases from 2001–2018 of 6.5% (95% CI 0.3% to 13.0%, p = 0.04) (Fig 1).

Primary and relapse VL patient characteristics

The flow of routinely-collected patient-level data from clinical records is shown in Fig 2. VL relapse as a proportion of all VL cases in patient-level data was consistent with programme monitoring data from Lankien for most of the study period, with VL relapse patients representing 6.2% of all VL patients in 2001–2003, 11.2% in 2013–2015 and 15.0% in 2016–2018. We compared this with programme data (Table 1) where we found 5.2%, 11.4% and 14.4% for the same time periods. During 2004–2012, the proportion of relapses in patient-level data from all sites (7.3%, 145/2,001) was lower than in programme data from Lankien (9.4%, 360/3,381), suggesting that relapse cases were more reliably diagnosed or recorded in Lankien and/ or that previously treated patients were more likely to return to Lankien for re-treatment and/ or that records missing from this period were more likely to be for readmitted patients.

VL patients (primary and relapse) were mainly children and adolescents: 26% of all patients were \leq 5 years old; 66% were \leq 16 years old. The median (IQR) age was 12 (5–22) years. There was a slight predominance of males (51.6%). These demographic characteristics were fairly constant over time (Table 2), albeit with an increase in the proportion of patients age \leq 16 years from 61.1% in 2001–2003 to 68.9% in 2016–2019 and a peak in the male-to-female ratio (55.0% male) in 2013–2015.

Anthropometric indices and haemoglobin levels were consistent with high prevalence of undernutrition and anaemia in the population throughout the study period (Table 2), with a



Table 1. Primary and relapse visceral leishmaniasis (VL) cases as proportions of all VL patients treated at Médecins Sans Frontières (MSF) centre in Lankien, South Sudan, 2001–2018 (from programme monitoring data).

https://doi.org/10.1371/journal.pntd.0010696.t001



Fig 1. Trend in relapse as a proportion of visceral leishmaniasis (VL) cases treated at Médecins Sans Frontières (MSF) centre in Lankien, South Sudan, 2001–2018 (from programme monitoring data)*. * Join-point analyses indicated that the best-fit (log linear) model had zero join points and an annual percent change in relapse as a proportion of all VL cases from 2001 to 2018 of 6.5% (95% CI 0.3% to 13.0%, p = 0.04).

https://doi.org/10.1371/journal.pntd.0010696.g001

mean (SD) weight-for-age Z-score of -1.6 (1.5) in the \leq 10-year-old age group, mean BMI-forage Z-score of -3.0 (1.1) in 11–19 year-olds, mean BMI of 15.7 (2.1) kg/m² in adults aged 20 years and older, and mean Hb of 9.2 (2.0) g/dL. Variation among primary VL patients suggested that anaemia and nutritional status of children \leq 10 years old improved from 2001– 2003 to 2013–2015 before worsening again in 2016–2018 and the nutritional status of older children and adults improved slightly overall. VL illness severity score was slightly higher in primary VL patients in 2001–2003 otherwise constant over time in both patient groups.

VL relapse patients did not differ from primary VL patients by median age (p = 0.76) or mean BMI for adults age 20+ years (Student's t test p = 0.77). There was a slightly higher proportion of males (53.9% vs. 51.4%, p = 0.03) and VL relapse patients age \leq 10y had slightly worse mean weight-for-age Z-score (-1.67 vs. -1.57, Student's t test p = 0.05) and slightly better mean BMI-for-age Z-score (-2.88 vs. -3.01, Student's t test p = 0.03).

Treatments and outcomes for primary VL and VL relapse patients

SSG/PM was the main treatment regimen from 2001–2018, used in 72.2% (13,772/19,063) of primary VL and 73.2% (1,509/2,061) of VL relapse cases (**Table 3**). SSG monotherapy was largely phased out by 2003. After 2012 AmBisome was used to treat 22.5% (2,245/9,986) of primary VL and 32.6% (470/1,440) of VL relapse cases from 2013–2018. Treatment default rates were consistently low over the years: 1.3% in 2001–2003; 1.6% in 2004–2012; 1.5% in 2015–2015; and 0.3% in 2016–2018.

Time from end of primary VL treatment to re-admission for VL relapse was constant over the study period (p = 0.12), with 66.9% (975/1,457) of all relapses for which time-to-relapse



⁺ Routinely-collected data were collated from several sources into a single spreadsheet hence, the need for de-duplication. This was done by dropping duplicates of a patient identifier comprising patient number + treatment site code + sex + date of admission

Fig 2. Flow of routinely-collected patient-level data from clinical records.

https://doi.org/10.1371/journal.pntd.0010696.g002

data were available occurring within 6 months. Nevertheless, 15.3% (223/1,457) of relapses occurred more than one year after primary VL treatment (Table 3).

As with anthropometric indices and haemoglobin levels, spleen sizes on admission improved slightly over the study period and the proportion of VL patients discharged without detectable splenomegaly increased from 88.6% (3,998/4,511) in 2014 to 99.2% (904/911) in 2018 (Table 4).

A small number (median 5, range 1–7) of HIV co-infected patients were detected each year from 2012 onwards, equivalent to an annual HIV co-infection rate in the years 2013–2018 of 0.1–0.4% (3.7% (7/189) in 2012) with no discernible trend. The proportion of relapse VL patients with a concomitant HIV infection (1.03%, 15/1,457) was more than six times higher than the HIV co-infection rate in primary VL patients (0.14%, 14/10,156), corresponding to an age-sex adjusted odds ratio of 6.58 (95% CI 3.15, 13.8) or 6.74 (3.22, 14.1) if patients with no HIV test result were considered to be negative. Similarly, the proportion of relapse VL patients with a confirmed concomitant tuberculosis (TB) infection (0.96%, 14/1,459) was more than four times higher than in primary VL patients (0.24%, 24/10,144), age-sex adjusted odds ratio 4.02 (95% CI 2.07, 7.81) or 4.09 (2.11, 7.93) if patients with no HIV test result were considered to be negative.

	2001-2003	2004-2012*	2013-2015	2016-2018	p-value [†]			
Primary VL	n = 7,223	n = 1,856	n = 6,503	n = 3,590				
Age (years), median (IQR)	12 (6-25)	12 (5-25)	10 (5–20)	13 (6-19)	< 0.001			
Sex (male), %	48.5%	54.3%	54.5%	50.1%	< 0.001			
Weight-for-age WHO Z-score, age ≤ 10 y, mean (SD)	-1.93 (1.36), n = 3,027	-1.58 (1.40), n = 792	-1.23 (1.51), n = 3,187	-1.55 (1.49), n = 1,219	< 0.001			
BMI-for-age WHO Z-score, age 11-19y, mean (SD)	-3.14 (1.08), n = 1,123	-3.01 (1.02), n = 299	-2.93 (1.13), n = 1,375	-2.95 (1.11), n = 1,303	0.001			
BMI, age >19y, mean (SD)	15.5 (2.0), n = 2,224	15.5 (2.0), n = 545	16.0 (2.1), n = 1,635	16.0 (2.1), n = 858	< 0.001			
Hb (g/dL), mean (SD)	7.8 (2.2), n = 1,095	8.9 (2.4), n = 802	9.5 (2.0), n = 6,462	9.2 (2.0), n = 3,527	< 0.001			
VL severity score (range 0–14), median (IQR)	3 (2–4), n = 1,048	2 (1–3), n = 1,029	2 (1-3), n = 6,503	2 (1–3), n = 3,591	< 0.001			
VL relapse	n = 476	n = 145	n = 821	n = 631				
Relapse as % of all VL patients	6.2%	7.3%	11.2%	15.0%	< 0.001			
Age (years)	12 (6-25)	13 (5–27)	10 (5–23)	13 (5-24)	0.01			
Sex (male)	46.7%	50.0%	59.6%	53.0%	< 0.001			
Weight-for-age WHO Z-score for age $\leq 10y$, mean (SD) -1.94 (1.39), n = 195	-1.78 (1.32), n = 61	-1.52 (1.52), n = 412	-1.69 (1.73), n = 233	0.09			
BMI-for-age WHO Z-score for age 11-19y, mean (SD)	-2.89 (1.12), n = 90	-2.94 (1.15), n = 18	-2.96 (1.22), n = 133	-2.87 (1.17), n = 160	0.93			
BMI, age >19y, mean (SD)	15.9 (2.1), n = 135	15.7 (2.1), n = 56	15.7 (2.2), n = 230	15.8 (1.9), n = 171	0.61			
Hb (g/dL), mean (SD)	9.1 (2.3), n = 195	8.6 (2.4), n = 99	9.0 (2.0), n = 819	8.8 (1.9), n = 593	0.12			
VL severity score (range 0–14), median (IQR)	2 (1-4), n = 163	2 (1-4), n = 101	2 (1–4), n = 821	2 (1-3), n = 630	0.14			

Table 2. Characteristics of primary visceral leishmaniasis (VL) and VL relapse patients treated at Médecins Sans Frontières (MSF) centres in South Sudan, 2001–2018 (patient-level data on admission).

* Data for 2004–2012 were combined because there were relatively low numbers of patients treated each year during this period, recording of patient-level data was not continuous and some records were lost as a result of attacks on MSF treatment centres

† Non-parametric K-sample test on the equality of medians for continuous variables; Chi-squared test for categorical variables

https://doi.org/10.1371/journal.pntd.0010696.t002

Discussion

Key finding and interpretation

Our analysis has shown that rates of VL relapse in South Sudan increased at a steady rate between 2001 and 2018, with a 2.4-fold overall increase in relapse as a percentage of all VL cases from 6% in 2001–2003 to 15% in 2016–2018.

An earlier analysis of MSF patient-level data (from 1999–2007) showed that age, sex, and malnutrition were not associated with risk of VL relapse, but a positive association with increased spleen size on admission and discharge was reported [19], whilst a slower rate of

•					
		2001-2003	2004-2012*	2013-2015	2016-2018
Primary VL treatment		n = 7,223	n = 1,856	n = 6,503	n = 3,590
	SSG	2,683 (37.2%)	140 (7.5%)	2 (0.0%)	0 (0.0%)
	SSG/PM	4,388 (60.8%)	1,655 (89.2%)	5,042 (78.1%)	2,687 (76.1%)
	AmBisome	130 (1.8%)	59 (3.2%)	1,401 (21.7%)	844 (23.9%)
	Other	20 (0.3%)	2 (0.1%)	9 (0.1%)	1 (0.0%)
Time to VL relapse (days)	median (IQR), n	139 (98–220), n = 248	118 (82–178), n = 80	126 (85–228), n = 777	132 (82–349), n = 352
	21-90 days	49 (19.8%)	25 (31.3%)	221 (28.4%)	105 (29.8%)
	91-180 days	115 (46.4%)	36 (45.0%)	312 (40.2%)	112 (31.8%)

Table 3. Treatment of visceral leishmaniasis (VL) patients and time to relapse at Médecins Sans Frontières (MSF) centres in South Sudan, 2001–2018 (from patient-level data).

* Data for 2004–2012 were combined because there were relatively low numbers of patients treated each year during this period, recording of patient-level data was not continuous and some records were lost as a result of attacks on MSF treatment centres

https://doi.org/10.1371/journal.pntd.0010696.t003

Primary VL	2014	2015	2016	2017	2018	p-value*
Admission (median (IQR))	2 (0–5), n = 4,119	4 (0–6), n = 978	2 (0–5), n = 1,284	2 (0-4), n = 1,562	2 (0–4), n = 686	0.008
Spleen 0 on discharge	89.0% (3,665/4,117)	92.9% (911/981)	92.1% (1,181/1,283)	93.8% (1,464/1,561)	99.4% (679/683)	< 0.001
VL relapse						
Admission (mean (SD))	4 (0–8), n = 397	4 (2–8), n = 410	4 (2–6), n = 142	4 (1–6), n = 228	4 (0–6), n = 228	0.05
Spleen 0 on discharge	84.5% (333/394)	94.4% (387/410)	83.1% (118/142)	94.7% (216/228)	98.7% (225/228)	< 0.001

Table 4. Spleen size (distance in centimetres from left costal margin to tip of spleen) on admission and discharge for primary and relapse visceral leishmaniais (VL) patients at Médecins Sans Frontières (MSF) centres in South Sudan, 2014–2018 (from patient-level data).

* Non-parametric test-for-trend across years for spleen size as continuous variable; Chi-squared test for spleen size 0 on discharge

https://doi.org/10.1371/journal.pntd.0010696.t004

regression of splenomegaly was positively associated with risk of VL relapse in India [20]. In our data, spleen size was recorded consistently only from 2013 onwards, therefore, we could not investigate trends over the whole study period. However, between 2014–2018 there was no substantial variation in median spleen sizes on admission, and the proportion of primary VL patients discharged with a normal spleen increased from 89% to 99%.

Low Hb levels were reported to be associated with increased risk of VL relapse in Brazil [21] and Georgia [22] but not in India [20]. We found that median HB levels on admission increased between 2001–2018, reflecting a general improvement in the nutritional status of older children and adults albeit from a very low baseline. The nutritional status of children \leq 10 years old improved from 2001–2003 to 2013–2015 but appeared to worsen again in 2016–2018, which gives cause for concern in terms of risks to child survival and to child growth and development. Co-infection with HIV and TB are strong risk factors for relapse, increasing the odds of relapse 7-fold and 4-fold, respectively, as seen in our data and elsewhere [23], but prevalence of these diseases has remained low (< 1%) in this population, with no observed trend. A VL severity score used operationally by MSF was slightly higher in primary VL patients in 2001–2003 but was otherwise static. Compliance to treatment was high throughout.

These relatively constant patient characteristics suggest that the substantial increase in incidence of VL relapse in this population must be explained by other factors. In the absence of evidence for changes in the susceptibility of the population or in the pathogenicity of *L. donovani* in East Africa, we must consider reduced effectiveness of treatment related to drug sensitivity. We must also exclude other factors as an explanation, including improvements in admission procedures or data recording leading to more re-admitted patients being identified correctly as VL relapse cases and improvements in security and health provision enabling more VL relapse cases to return for treatment.

We would expect better access to VL treatment to apply to all cases, primary and relapse therefore, this would be unlikely to bias our calculation of the latter as a proportion of all cases. Similarly, we cannot attribute the increase in VL relapse to larger numbers of older patients who had primary VL further back in time, because patient age (71% under 20 years old) and time-to-relapse (67% within 6 months) were constant in our data. Neither did our data indicate higher apparent VL relapse rates related to easier access to treatment centres during the relatively peaceful years between the 2005 peace treaty and the outbreak of civil war in 2013, although total case numbers were lower during this period than in other years. A higher proportion of relapses can also be expected to be observed in the year following a relatively higher number of VL cases because, in the second year, the number of new VL cases decreases and the number of relapses proportionally increases. We can discern this in our programme data for 2004 (16% VL relapse) following the 2002/2003 outbreak, 2011/2012 (11%) following the 2010 outbreak, and 2015 (28%) following the 2014 outbreak. The steady upward trend in VL relapse cases superimposed on these and other short-term fluctuations in our data was supported by moderate statistical evidence, and future analysis using a longer time series might discern correlation with background VL incidence, seasonality, and other ecological factors.

Implications for policy and practice

If resistance has emerged for paromomycin, a drug for which resistance is easily induced *in vitro* [24], and/or for SSG, for which widespread resistance was demonstrated in India after decades of use, then treatment with 17 days SSG/PM will leave many patients parasitaemic despite a good initial clinical response, who will then be at increased risk of relapse. This will further increase the probability of development of resistant strains for either component of the combined regimen. There is also some evidence that the current regimen of 30mg/kg AmBisome is not fully effective in East African VL [25], with many patients requiring extended treatment. Although this regimen is recommended by WHO as a second-line treatment of VL caused by *L. donovani* in East Africa, evidence for this regimen is lacking. The regimen of 30mg/kg was decided somewhat arbitrarily, based on studies for treatment of VL caused by *L. infantum* in the Mediterranean [26,27]. Later studies with *L. donovani* in East Africa showed high failure rates in dose-finding studies which only went up to 21mg/kg [28].

High retreatment rates have not been reported from other East African countries [13,14]. However, the SSG/PM combination regimen has only been rolled out in other countries more recently (after 2010), whereas it was introduced in the MSF programmes in South Sudan in 2002–2003. *L. donovani* parasites in our study area have therefore been exposed to the combination regimen for at least 7 years longer than in other countries, indicating that our data from South Sudan might provide early warning for the emergence of parasite strains with decreased drug sensitivity.

Limitations

The main limitation of our study is that routinely-collected demographic and clinical data from resource-poor and conflict-affected settings will be subject to errors of measurement, recording and classification, although these are more likely to be random than systematic and will therefore tend to bias estimates towards the null. Treatments were standardised (SSG17/ PM17 for primary VL, SSG30/PM17 for VL relapse, 6 x 5mg/kg AmBisome for both) and protocols implemented to ensure adherence and minimise default, but patient-level recording of the precise treatment received (dose and duration), test-of-cure and whether patients were readmitted for VL relapse are subject to error. VL relapse patients were usually issued with new patient numbers because of the very similar names used by ethnic groups in the region and lack of fixed home addresses or other identifiers that could be used to link primary to relapse cases, while reliability of patient recall of previous treatment is unknown. This means that some VL relapse cases may have been misclassified as primary VL or vice versa. Differentiation between relapse and reinfection can only be established by isoenzyme characterisation, which is not possible in programme settings. However, very limited research (in VL patients with HIV co-infection) indicates that cases presenting with a history of previous VL episodes are much more likely to be relapse than re-infection [29]. Another limitation of our analysis is missing data, particularly in years when patient-level data collection was interrupted (2005– 2008 and 2011). However, aggregate programme monitoring data was collected during these periods and we think it is sufficiently reliable for monitoring long-term trends. We also noted that VL relapse as a proportion of all cases was similar in programme and patient-level data and both data sources yielded similar trend estimates. The annual percentage change in the best fitting model was supported by marginal statistical evidence and the confidence interval

was quite wide, reflecting substantial year-on-year variation in VL relapse rates. However, the underlying data showing VL relapse increasing from 6% of all VL patients in 2001–2003 to 11% in 2013–2015 and 15% in 2016–2018 gives us a fair degree of confidence that there is a real effect, and the clinical significance of this increase in VL relapse is high.

Suggestions for the future

Given the limitations of the current retrospective study, MSF is planning a prospective observational study to assess the incidence of relapse (and PKDL) after treatment for primary VL in South Sudan in which patients will be followed up for 12 months. The results of this prospective study will indicate whether drug sensitivity monitoring of anti-leishmanial drugs being used East Africa needs to be considered and will give an indication of the optimal duration of follow-up after treatment to detect the majority of relapses.

The high proportion of relapses presenting to our clinic indicate the importance of encouraging patients to return when symptons re-occur, and of the necessity of close populationlevel monitoring by national control programs in the region of all VL treatment regimen. However, whether strategies that ensure 6-month or 12-month follow-up can be translated from stable settings such as India or relatively well-resourced effectiveness studies to the South Sudanese context is doubtful at present [30]. Genotyping of parasite isolates from different VL patient categories (primary/relapse, cured/failed) might also reveal predisposing factors to unfavourable treatment outcomes.

Currently under study in 4 East African countries is a 14-day combination regimen of PM (20 mg/kg) in combination with miltefosine (allometric dosing), an oral drug. This combination represents an increased dose of PM, while MF is a drug not previously used in South Sudan. This regimen may provide a viable alternative to SSG/PM. As miltefosine is teratogenic, precautions will need to be taken when administering it to women of child-bearing age and to prevent administration during pregnancy. This study will also generate new data on the efficacy and 6-month relapse rates of SSG/PM, the control arm.

Conclusion

We have found that increasing incidence of VL relapse in South Sudan does not appear to be explained by changes in patient characteristics or artefact and therefore the effectiveness of current and proposed VL treatments in this setting require urgent and careful evaluation. Drug resistance emerging after decades of SSG/PM combination therapy is a worrying prospect as no alternative treatments are yet available. This underscores the pressing need for new chemical entities that will enable safe and highly effective short-course oral treatments for VL. Until these are available, the new PM/MF regimen that is currently under study can be considered as an alternative treatment option.

Supporting information

S1 Table. Médecins Sans Frontières (MSF) visceral leishmaniasis (VL) programme data from Lankien and patient-level data from all MSF sites in South Sudan, 2001–2018. (PDF)

Author Contributions

Conceptualization: Margriet den Boer, Koert Ritmeijer.

Data curation: Gabriel Naylor-Leyland, Simon M. Collin.

Formal analysis: Gabriel Naylor-Leyland, Simon M. Collin.

Supervision: Margriet den Boer, Koert Ritmeijer.

- Writing original draft: Gabriel Naylor-Leyland, Simon M. Collin, Francis Gatluak, Margriet den Boer, Fabiana Alves, Abdul Wasay Mullahzada, Koert Ritmeijer.
- Writing review & editing: Gabriel Naylor-Leyland, Simon M. Collin, Francis Gatluak, Margriet den Boer, Fabiana Alves, Abdul Wasay Mullahzada, Koert Ritmeijer.

References

- 1. Bhattacharya SK, Dash AP: Elimination of Kala-Azar from the Southeast Asia Region. *Am J Trop Med Hyg* 2017, 96(4):802–804. https://doi.org/10.4269/ajtmh.16-0279 PMID: 28115678
- 2. Al-Salem W, Herricks JR, Hotez PJ: A review of visceral leishmaniasis during the conflict in South Sudan and the consequences for East African countries. *Parasit Vectors* 2016, 9(1):460. <u>https://doi.org/10.1186/s13071-016-1743-7 PMID: 27549162</u>
- Dahl EH, Hamdan HM, Mabrouk L, Matendechero SH, Mengistie TB, Elhag MS et al: Control of visceral leishmaniasis in East Africa: fragile progress, new threats. *BMJ Glob Health* 2021, 6(8). https://doi.org/ 10.1136/bmjgh-2021-006835 PMID: 34413075
- 4. Sunyoto T, Potet J, den Boer M, Ritmeijer K, Postigo JAR, Ravinetto R et al: Exploring global and country-level barriers to an effective supply of leishmaniasis medicines and diagnostics in eastern Africa: a qualitative study. *BMJ Open* 2019, 9(5):e029141. https://doi.org/10.1136/bmjopen-2019-029141 PMID: 31152044
- 5. Sundar S: Drug resistance in Indian visceral leishmaniasis. *Trop Med Int Health* 2001, 6(11):849–854. https://doi.org/10.1046/j.1365-3156.2001.00778.x PMID: 11703838
- Hendrickx S, Guerin PJ, Caljon G, Croft SL, Maes L: Evaluating drug resistance in visceral leishmaniasis: the challenges. *Parasitology* 2018, 145(4):453–463. https://doi.org/10.1017/S0031182016002031 PMID: 27866478
- Melaku Y, Collin SM, Keus K, Gatluak F, Ritmeijer K, Davidson RN: Treatment of kala-azar in southern Sudan using a 17-day regimen of sodium stibogluconate combined with paromomycin: a retrospective comparison with 30-day sodium stibogluconate monotherapy. *Am J Trop Med Hyg* 2007, 77(1):89–94. PMID: 17620635
- Seaman J, Pryce D, Sondorp HE, Moody A, Bryceson AD, Davidson RN: Epidemic visceral leishmaniasis in Sudan: a randomized trial of aminosidine plus sodium stibogluconate versus sodium stibogluconate alone. *J Infect Dis* 1993, 168(3):715–720. <u>https://doi.org/10.1093/infdis/168.3.715</u> PMID: 8394861
- Musa A, Khalil E, Hailu A, Olobo J, Balasegaram M, Omollo R et al: Sodium stibogluconate (SSG) & paromomycin combination compared to SSG for visceral leishmaniasis in East Africa: a randomised controlled trial. *PLoS Negl Trop Dis* 2012, 6(6):e1674. https://doi.org/10.1371/journal.pntd.0001674 PMID: 22724029
- WHO Expert Committee on the Control of the Leishmaniases: Control of the leishmaniases: report of a meeting of the WHO Expert Committee on the Control of Leishmaniases, Geneva, 22–26 March 2010.
 In. Geneva: World Health Organization; 2010.
- 11. Ministry of Health: Guidelines for diagnosis, treatment and prevention of visceral leishmaniasis in South Sudan, Juba, South Sudan; 2011.
- Seaman J, Boer C, Wilkinson R, de Jong J, de Wilde E, Sondorp E et al: Liposomal amphotericin B (AmBisome) in the treatment of complicated kala-azar under field conditions. *Clin Infect Dis* 1995, 21 (1):188–193. https://doi.org/10.1093/clinids/21.1.188 PMID: 7578729
- **13.** Atia AM, Mumina A, Tayler-Smith K, Boulle P, Alcoba G, Elhag MS et al: Sodium stibogluconate and paromomycin for treating visceral leishmaniasis under routine conditions in eastern Sudan. *Trop Med Int Health* 2015, 20(12):1674–1684. https://doi.org/10.1111/tmi.12603 PMID: 26427033
- Kimutai R, Musa AM, Njoroge S, Omollo R, Alves F, Hailu A et al: Safety and Effectiveness of Sodium Stibogluconate and Paromomycin Combination for the Treatment of Visceral Leishmaniasis in Eastern Africa: Results from a Pharmacovigilance Programme. *Clin Drug Investig* 2017, 37(3):259–272. <u>https:// doi.org/10.1007/s40261-016-0481-0</u> PMID: 28066878
- World Health Organization Division of Control of Tropical Diseases: Manual on visceral leishmaniasis control. In. Geneva: World Health Organization; 1996.

- Balasegaram M, Ritmeijer K, Lima MA, Burza S, Ortiz Genovese G, Milani B et al: Liposomal amphotericin B as a treatment for human leishmaniasis. *Expert opinion on emerging drugs* 2012, 17(4):493–510. https://doi.org/10.1517/14728214.2012.748036 PMID: 23167833
- World Health Organization: Control of the leishmaniases: WHO TRS N°949. In: Report of a meeting of the WHO Expert Committee on the Control of Leishmaniases, Geneva, 22–26 March 2010. Geneva: World Health Organization; 2010.
- Kim HJ, Fay MP, Feuer EJ, Midthune DN: Permutation tests for joinpoint regression with applications to cancer rates. *Stat Med* 2000, 19(3):335–351. https://doi.org/10.1002/(sici)1097-0258(20000215) 19:3<335::aid-sim336>3.0.co;2-z PMID: 10649300
- Gorski S, Collin SM, Ritmeijer K, Keus K, Gatluak F, Mueller M et al: Visceral leishmaniasis relapse in Southern Sudan (1999–2007): a retrospective study of risk factors and trends. *PLoS Negl Trop Dis* 2010, 4(6):e705. https://doi.org/10.1371/journal.pntd.0000705 PMID: 20544032
- Burza S, Sinha PK, Mahajan R, Lima MA, Mitra G, Verma N et al: Risk factors for visceral leishmaniasis relapse in immunocompetent patients following treatment with 20 mg/kg liposomal amphotericin B (Ambisome) in Bihar, India. *PLoS Negl Trop Dis* 2014, 8(1):e2536. https://doi.org/10.1371/journal.pntd. 0002536 PMID: 24392166
- Simao JC, Victoria C, Fortaleza C: Predictors of relapse of visceral leishmaniasis in inner Sao Paulo State, Brazil. Int J Infect Dis 2020, 95:44–49. https://doi.org/10.1016/j.ijid.2020.02.028 PMID: 32088340
- Kajaia M, Morse DL, Kamkamidze G, Butsashvili M, Chubabria G, Zenaishvili O et al: Risk factors for relapse of visceral leishmaniasis in Georgia. *Trop Med Int Health* 2011, 16(2):186–192. https://doi.org/ 10.1111/j.1365-3156.2010.02694.x PMID: 21143353
- Gidey K, Belay D, Hailu BY, Kassa TD, Niriayo YL: Visceral Leishmaniasis Treatment Outcome and Associated Factors in Northern Ethiopia. *Biomed Res Int* 2019, 2019:3513957. https://doi.org/10.1155/ 2019/3513957 PMID: 31531350
- 24. Davidson RN, den Boer M, Ritmeijer K: Paromomycin. *Trans R Soc Trop Med Hyg* 2009, 103(7):653–660.
- Mueller M, Ritmeijer K, Balasegaram M, Koummuki Y, Santana MR, Davidson R: Unresponsiveness to AmBisome in some Sudanese patients with kala-azar. *Trans R Soc Trop Med Hyg* 2007, 101(1):19– 24. https://doi.org/10.1016/j.trstmh.2006.02.005 PMID: 16730363
- Davidson RN, di Martino L, Gradoni L, Giacchino R, Gaeta GB, Pempinello R et al: Short-course treatment of visceral leishmaniasis with liposomal amphotericin B (AmBisome). *Clin Infect Dis* 1996, 22 (6):938–943. https://doi.org/10.1093/clinids/22.6.938 PMID: 8783690
- Davidson RN, Di Martino L, Gradoni L, Giacchino R, Russo R, Gaeta GB et al: Liposomal amphotericin B (AmBisome) in Mediterranean visceral leishmaniasis: a multi-centre trial. *Q J Med* 1994, 87(2):75– 81. PMID: 8153291
- Khalil EA, Weldegebreal T, Younis BM, Omollo R, Musa AM, Hailu W et al: Safety and efficacy of single dose versus multiple doses of AmBisome for treatment of visceral leishmaniasis in eastern Africa: a randomised trial. *PLoS Negl Trop Dis* 2014, 8(1):e2613. https://doi.org/10.1371/journal.pntd.0002613 PMID: 24454970
- Morales MA, Cruz I, Rubio JM, Chicharro C, Cañavate C, Laguna F et al: Relapses versus reinfections in patients coinfected with Leishmania infantum and human immunodeficiency virus type 1. J Infect Dis 2002, 185(10):1533–1537. https://doi.org/10.1086/340219 PMID: 11992294
- Goyal V, Burza S, Pandey K, Singh SN, Singh RS, Strub-Wourgaft N et al: Field effectiveness of new visceral leishmaniasis regimens after 1 year following treatment within public health facilities in Bihar, India. *PLoS Negl Trop Dis* 2019, 13(9):e0007726. https://doi.org/10.1371/journal.pntd.0007726 PMID: 31557162